Author's response to reviews

Title: Is Canada ready for patient accessible electronic health records? A national scan.

Authors:

Sara Urowitz (sara.urowitz@uhn.on.ca)
David Wiljer (david.wiljer@uhn.on.ca)
Emma Apatu (emma.apatu@uhn.on.ca)
Gunther Eysenbach (geysenba@ehealthinnovation.org)
Claudette DeLenardo (claudette.delenardo@grhosp.on.ca)
Tamara Harth (tamara.harth@sunnybrook.ca)
Howard Pai (hpai@bccancer.bc.ca)
Kevin J Leonard (k.leonard@utoronto.ca)

Version: 6  Date: 21 July 2008

Author's response to reviews: see over
July 21, 2008

Dr. Anastasios Koutsos, Assistant Editor  
BMC-series journals  
Tel: +44 (0)20 7631 9921  
Facsimile: +44 (0)20 7631 9923  
e-mail: editorial@biomedcentral.com  
Web: http://www.biomedcentral.com/

Dear Dr. Koutsos:

We have now completed the final proof reading of our manuscript and we are submitting a final version for publication in the BMC-series journals.

Sincerely,
Sara

Sara Urowitz, MSW, PhD  
Manager, Educational Informatics and  
Caring Voices Professional Lead (www.caringvoices.ca)  
Oncology Education  
Princess Margaret Hospital/University Health Network  
610 University Ave. (5-312)  
Toronto, ON M5G 2M9  
(p) 416-946-4501 ext 3810  
(f) 416-946-4442  
sara.urowitz@uhn.on.ca